Author's response to reviews

Title: Atypical haemolytic uraemic syndrome presenting initially as suspected meningococcal disease

Authors:

Siddharthan Sivamurthy (ssivamurthy@yahoo.co.uk)
John D Mooney (john.mooney@ferndown.nhs.uk)
Tom D Kenny (Tom.Kenny@ferndown.nhs.uk)

Version: 2 Date: 9 March 2007

Author's response to reviews:

As requested to conform to journal style:
Title page subheading removed
Title now ending in ': A Case Report.'
Written patient consent confirmed as received in Acknowledgement section.